Author's response to reviews

Title: Dying comfortably in very old age with or without dementia in different care settings - a representative "older old" population study

Authors:

Jane Fleming (jane.fleming@phpc.cam.ac.uk)
Rowan Calloway (rowancalloway@gmail.com)
Anouk Perrels (anoukperrels@hotmail.com)
Morag Farquhar (M.Farquhar@uea.ac.uk)
Stephen Barclay (sigb2@medschl.cam.ac.uk)
Carol Brayne (cb105@medschl.cam.ac.uk)
Cambridge City over-75s Cohort (CC75C) study collaboration (fjf22@cam.ac.uk)

Version: 3 Date: 29 Aug 2017

Author’s response to reviews:

Our response to reviewers previously submitted has now been removed as requested in 14 July 2017 e-mail from the editorial team.

Today's revised version (29 August) differs from the last uploaded version (22 August) only in:

- replacement of links in Funders and Acknowledgements sections that had changed due to updates to the study website since this paper was first submitted (already noticed and e-mailed 23 August, but not uploaded as no 'revise' link available then)

- rewording Author Contribution section as requested in your 24 August e-mail ("read and approved" instead of "agreed")

- one word cut from Abstract to fit 350 word limit ("whether home or care home" instead of "whether at home or care home")